Author's response to reviews

Title: Compound K, a metabolite of ginseng saponin, induces apoptosis via caspase-8-dependent pathway in HL-60 human leukemia cells

Authors:

Sung-Hee Cho (skstlswltjs@nate.com)
Kyung-Sook Chung (adella76@hanmail.net)
Jung-Hye Choi (apoptosis00@msn.com)
Dong-Hyun Kim (dhkim@khu.ac.kr)
Kyung-Tae Lee (ktlee@khu.ac.kr)

Version: 4 Date: 11 December 2009

Author's response to reviews: see over
Dear Editor

We appreciate reviewers’ valuable comments and have revised our manuscript accordingly. We hope that the revised form is now adequate for publication in BMC cancer.

Comments of Reviewer and Response:

1. Figure titles: Each figure should have a title one sentence in length. All other information should be included in the figure legend. Figure legends should be no longer than 250 words.
   -> Figure legends has been changed as the reviewer suggested in the revised manuscript.

2. Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.
   -> Highlighting/tracking has been removed in the revised manuscript.

3. Figure citations: Please be consistent when referring to figures, using either ‘Figure’ or ‘Fig’ throughout the text and figure legends.
   -> Figure citations has been consistent using ‘Figure’ in the revised manuscript.

4. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
   -> It has been checked in the revised manuscript.

Thank you very much for your advices.

Sincerely yours,

Kyung-Tae Lee, Ph.D.
Department of Pharmaceutical Biochemistry, College of Pharmacy
Kyung-Hee University
Dongdaemun-Ku, Hoegi-Dong 130-701, Seoul
Korea
E-mail: ktlee@khu.ac.kr
Tel: +82-2-9610860, FAX: +82-2-9620860